Author’s response to reviews

Title: Effect of intraocular anti-VEGF on cystoid macular edema associated with Henoch- Schonlein purpura

Authors:

dongyan pan (campus@126.com)
Xiao Cui (cuixiai1024@163.com)
Weiyu Zhu (wangfangzouyue@163.com)
Haifeng Qin (qhf0214@126.com)
Jin-hui Wu (wjh2042@aliyun.com)
qing li (154256398@qq.com)
bing xu (insanes@163.com)

Version: 2 Date: 15 Feb 2020

Author’s response to reviews:

Dear editor,

Thank you for your letter and the comments, we have made the required corrections to the manuscript.

1. Authors Contributions
   We have added the authors [Qing Li, Bing Xu] to the Authors’ Contributions section.

2. Timeline figure
   We have uploaded the Timeline as figure 4.

3. Discussion and Conclusion sections
   We have combined the Discussion and Conclusion sections into one section titled "Discussions and conclusion"

4. Clean manuscript
   We have cleaned all tracked changes, comments, highlights, strikethroughs and colored text.

Yours sincerely,
Jinhui Wu